Author's response to reviews

Title: Knowledge in process? Exploring barriers between epidemiological research and local health policy development.

Authors:

Joyce de Goede (j.degoede@uvt.nl)  
Kim Putters (putters@bmg.eur.nl)  
Tom van der Grinten (vandergrinten@bmg.eur.nl)  
Hans AM van Oers (hans.van.oers@rivm.nl)

Version: 2 Date: 2 October 2009

Author's response to reviews:

On 2 oktober I resubmitted mu paper because of some typing errors.